Author’s response to reviews

Title: Nocardia farcinica lung infection in a patient with cystic fibrosis: a case report

Authors:

Fadi Bittar (bittar_fadi@hotmail.com)
Nathalie Stremler (nathalie.stremler@ap-hm.fr)
Jean-Pierre Audié (jean-pierre.audie@ch-martiques.fr)
Jean-Christophe Dubus (jean-christophe.dubus@ap-hm.fr)
Jacques Sarles (jacques.sarles@ap-hm.fr)
Didier Raoult (didier.raoult@gmail.com)
Jean-Marc Rolain (jean-marc.rolain@univmed.fr)

Version: 2 Date: 27 January 2010

Author’s response to reviews: see over
Marseille, 27/01/2010

Journal of Medical Case Reports

Dear Editor,

Please find enclosed our revised manuscript entitled «Nocardia farcinica lung infection in a patient with cystic fibrosis: a case report». All authors have seen and approved the manuscript.

Formatting changes to be made:
Major revisions (We require the author to make these changes)
Ethnicity: Please provide the ethnicity and nationality of the patient described in the case study.

These have been provided in the revised manuscript.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title page: Please remove the `received' and `accepted' date sections from the manuscript.

These have been removed in the revised manuscript

Check changes: Please turn off all check changes, comments and formatting notes from the manuscript.

All changes, comments and formatting notes have been checked and turned off in the revised manuscript.

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors. For example, references 5 and 13 currently use `et al'.

These have been corrected in the revised manuscript.
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

No typographical errors have been found.

I hope that the revised manuscript is now convenient.

Sincerely yours.

Pr ROLAIN Jean-Marc